Author’s response to reviews

Title: Tracking physical activity using smart phone apps: Assessing the ability of a current app and systematically collecting patient recommendations for future development

Authors:

Jennifer Murphy (j.murphy@imperial.ac.uk)
Trishna Uttamlal (phd17tu@mail.wbs.ac.uk)
Kelly Schmidtkke (K.Schmidtke@mmu.ac.uk;kelly.ann.schmidtke@icloud.com)
Ivo Vlaev (Ivo.Vlaev@wbs.ac.uk)
David Taylor (david.taylor@imperial.ac.uk)
Mian Ahmad (mian.ahmad@imperial.ac.uk)
Susan Alsters (s.alsters@imperial.ac.uk)
Sanjay Purkayastha (s.purkayastha@imperial.ac.uk)
Samantha Scholtz (Samantha.Scholtz2@imperial.nhs.uk)
Ramin Ramezani (raminr@ucla.edu)
Ahmed Ahmed (Ahmed.Ahmed@imperial.nhs.uk)
Harvinder Chahal (h.chahal@imperial.ac.uk)
Ara Darzi (a.darzi@imperial.ac.uk)
Alexandra Blakemore (a.blakemore@imperial.ac.uk)

Version: 4 Date: 03 Jan 2020

Author’s response to reviews:

Revision Request 1: Please state in the cover letter whether the images depicted in additional file 1 are your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.

Should you require an alternative source you may wish to try Wikimedia Commons: http://commons.wikimedia.org/wiki/Main_Page
Authors response: The image of the Moves app has been removed. The images of the WLCompanion app remain, as this app was developed by Imperial College London by co-authors we have the rights to use it in this publication. A revised version of additional file 1 is attached in this submission.

Revision Request 2: Please remove the, response to reviewers and change in authorship form as they are no longer needed at this stage.

Authors response: Response to the reviewers and change in authorship form have been removed from the attached files. The responses to revisions requests are input into the designated box.

Revision Request 3. *As a reminder* please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Authors response: The manuscript is in its final form for submission (no track changes or highlighting).

Further Request 1: Please clarify if Dr. Alister, SI (manuscript) and Susan I Alsters (system) is the same author. If so, please correct so they are consistent with each other.

Authors response: We have verified the correct name with the author as "Suzanne Alsters"